Author's response to reviews

Title: Novel SYBR-based duplex qPCR for the detection of gene dosage: Detection of an APC large deletion in a Familial adenomatous polyposis patient with unusual phenotype.

Authors:

Giovana T Torrezan (gitorrezan@hotmail.com)
Felipe C C Silva (felipe.silva@accamargo.org.br)
Ana C V Krepischi (ana.krepischi@gmail.com)
Erika M M Santos (emonteiro@accamargo.org.br)
Benedito M Rossi (rossibm@gmail.com)
Dirce M Carraro (dirce.carraro@accamargo.org.br)

Version: 2 Date: 1 February 2012

Author's response to reviews:

The manuscript was revised following the Editorial Office requests and the following changes were performed:

1. The following statement was added to the Methods section of the manuscript: "This study was performed in compliance with the Helsinki Declaration and was approved by the ethics committee of A. C. Camargo Hospital (approval number: 1169/08-B)."

2. The abstract was restructured to contain a methods section.